Author's response to reviews

Title: Fulfillment of the Brazilian Agenda of Priorities in Health Research

Authors:

Leonor Maria Pacheco Santos (leopac@unb.br)
Erly Catarina Moura (erly.moura@saude.gov.br)
Rita de Cássia Barradas Barata (rita.barradasbarata@gmail.com)
Suzanne Jacob Serruya (serruyas@who.int)
Marcia Luz da Motta (marcia.motta@saude.gov.br)
Flavia Tavares Silva Elias (flavia.elias@saude.gov.br)
Antonia Angulo-Tuesta (antoniaangulo@unb.br)
Ana Patricia de Paula (apatriciapaula@gmail.com)
Gilvania de Melo (gilvania.melo@saude.gov.br)
Reinaldo Guimarães (reinaldoguimaraes47@gmail.com)
Carlos Augusto Grabois Gadelha (carlos.gadelha@saude.gov.br)

Version: 11  Date: 27 August 2011

Author's response to reviews:

Dear Sirs

I am trying once more to change my files in the system.

There were some typos that I changed in this new version, but the worse of all were the Figure Legends.

I will attach the last revised version (Proof 5.doc). At the end, the Figure Legends as they should be. The two sentences shown below should be added to both Figures, but it is not really "Legend" but rather "Footnote"

"Source: data summaries were calculated by the authors based on information taken from: http://www.saude.gov.br/sisct and http://www.saude.gov.br/pesquisasaude, captured on Jan 21, 2011."

Best

Leonor